CSE 326: Data Structures

Spring 1996

Question naire

This questionnaire has t w o pages. It is not a test, and will not b e graded. The goal is

for me to get a go o d understandin g of what y ou kno w from previous courses and what y ou

ha v e y et to learn.

Name:

Electronic address:

Ma jor: CS or CE or other (circle one; if \other", please sp ecify)

Quarter T aken L e ctur er Pr o gr amming L anguage

CSE 142 (Prog. I)

CSE 143 (Prog. I I)

CSE 321 (Disc. Struct.)

Ha v e y ou used Unix? If so, in whic h course?

Ha v e y ou used g++? If so, in whic h course?

Ha v e y ou learned

� link ed lists?

� stac ks?

� queues?

� trees?

� preorder tra v ersal?

� inorder tra v ersal?

� p ostorder tra v ersal?

� big \O" notation?

� ho w to eliminate recursion using stac ks?

If y ou learned it, what is the di�erence b et w ee n a list and a set?

If y ou learned it, brie y describ e (in English) ho w preorder tra v ersal w orks.

If y ou learned it, list the stac k op erations and brie y describ e what eac h one do es.

If y ou learned it, list the queue op erations and brie y describ e what eac h one do es.

Is 47n

2

+ 93n log n = O (n

3

)?

Is 47n

2

+ 93n log n = O (n

2

)?

Is 47n

2

+ 93n log n = O (n log n)?